May 16, 2021

Dear editor,

Please find attached files of revised manuscript in word format

**Title:** Choice of operative methods for pancreaticojejunostomy and multivariable study of pancreatic leakage in PD

**Author:** Hui Liang¹, Jianguo Wu², Fei Wang³, Boxuan Chen⁴, Shitian Zou⁵, Cong Wang⁶, Shuaiwu Luo⁷

**Name of Journal:** World Journal of Gastrointestinal surgery

**Manuscript NO:** 65067

First of all, thank you for your careful guidance of this article. Revision has been made according to the suggestions of the reviewer:

**Reviewer: 06058930**
Pancreatoduodenectomy is involved many organs, and its operation is complex and the scope of operation is large, which can cause great trauma in patients and has the high occurrence rate of complications. Not all patients can adopt the only pancreaticojejunostomy, among which, the pancreaticojejunostomy and the bundled pancreaticojejunostomy are commonly used. There are still some differences about what kind of pancreaticojejunostomy is more favorable to patients. In this study, the authors reviewed the curative effect of pancreaticojejunostomy and binding pancreaticojejunostomy in pancreatoduodenectomy and analyzed the multiple factors of postoperative pancreatic leakage. This study overall is very interesting. The methods are reasonable and results are well discussed.

Comments: 1. The manuscript requires a minor language editing. Please take attention about the spellings. And some blanks between the words are missing, such as: Abstract, methods “operation, between which, the” should be “operation, between which, the”, Abstract, results “differences of” should be “differences of” etc. Please check the manuscript carefully again. 2. It seems the figure is same as the table, please delete it. Or refer it in the main text.

**Reviewer: 06058923**
This is an interesting study of multivariable study of pancreatic leakage in pancreaticoduodenectomy. The manuscript is very well written. The research objectives was achieved by the experiments used in this study. A minor editing is required for the manuscript before publishing online.

After receiving the comments, we read the article carefully and found some small loopholes in the language of the article and made modifications. We have revised the problem of the picture and added a reference to the text.

Thank you again for publishing our manuscript in the World Journal of Clinical Cases.

Sincerely Yours,

Jianguo Wu